Author's response to reviews

Title: CD155/PVR plays a key role in cell motility during tumor cell invasion and migration

Authors:
Kevin E Sloan (kevin.sloan@tufts.edu)
Brenda K’ Eustace (brenda.eustace@tufts.edu)
Jean K Stewart (jean.stewart@tufts.edu)
Carol Zehetmeier (c.zehetmeier@xerion-pharma.com)
Claudia Torella (c.torella@xerion-pharma.com)
Marina Simeone (marina.simeone@tufts.edu)
Jennifer E Roy (royj@helix.mgh.harvard.edu)
Christine Unger (c.unger@xerion-pharma.com)
David N Louis (dlouis@partners.org)
Leodevico L Ilag (l.ilag@xerion-pharma.com)
Daniel G Jay (daniel.jay@tufts.edu)

Version: 4 Date: 5 October 2004

Author's response to reviews:

10/5/04

Changes to manuscript:

* Changed wording in Results, p14, per reviewer suggestion: "Since normal and cancerous tissue samples were not collected from the same patient, we cannot determine if elevated CD155 expression correlates with tumorigenesis..."

* Running title - removed the running title from the title page.

* Keywords section - removed the keywords section from the title page.

* Manuscript- removed the header from the manuscript pages.